Author's response to reviews

Title: The Development of the Canberra Symptom Scorecard: A Tool to Monitor the Physical Symptoms of Patients with Advanced Tumours.

Authors:

Margherita J Barresi (margherita.barresi@defence.gov.au)
Dr Bruce Shadbolt (bruce.shadbolt@act.gov.au)
Don Byrne (don.byrne@anu.edu.au)
Robin Stuart-Harris (robin.stuart-harris@act.gov.au)

Version: 2 Date: 5 Nov 2003

Please see accompanying Word document entitled CSS Responses to Reviewers 031103